Author's response to reviews

Title: Humoral immune response to MUC5AC in patients with colorectal polyps and colorectal carcinoma

Authors:

Belma Kocer (belmak@lycos.com)
John McKolanis (mckolani+pitt.edu)
Atilla Soran (asoran@magee.edu)

Version: 2 Date: 23 August 2005

Dear Editor

According to your comments for abstract below changes are made:
1-We sub-divide your abstract into Background, Methods, Results and Conclusions subsections.
2-A sentence was added to the background section in the abstract: MUC5AC is as secreted mucin aberrantly expressed by colorectal polyps and carcinoma. It has been hypothesized that aberrant expression of MUC5AC in colorectal carcinoma tissues increased the overall survival of patients with colorectal carcinoma.
3-In order to make the statement clear conclusion was changed as "Decreased survival in colorectal carcinoma patients with MUC5AC antibody positivity may be due to a decreased in the MUC5AC expression in tumor tissues of surviving carcinoma patients."